Author's response to reviews

Title: Frontalis Suspension Surgery to Treat Patients with Essential Blepharospasm and Apraxia of Eyelid Opening - Technique and Results -

Authors:

Chrisanthi Karapantzou (chrisanthis24@yahoo.gr)
Dirk Dressler (Dressler.Dirk@mh-hannover.de)
Saskia Rohrbach (saskia.rohrbach@googlemail.com)
Rainer Laskawi (rlaskaw@gwdg.de)

Version: 2 Date: 4 September 2014

Author's response to reviews:

Dear Mr. Failma, Dear Ladies and Gentlemen,

According to your “Formatting-Email” we resubmit our paper entitled “Frontalis Suspension Surgery to Treat Patients with Essential Blepharospasm and Apraxia of Eyelid Opening
- Technique and Results - “.

We corrected the paper and integrated the needed changes you listed:

(a) The contribution of each author is mentioned at the end of the manuscript

(b) Patient’s identity: Unfortunately it is not possible to cover the eyes of the patients because this is important for demonstration of the surgical effect. But, as wanted, we modified all figures and focused on the affected areas. As we told you in our initial cover letter all patient portraited in the figures gave their informed consent for publication of figures in which they are recognizable.

(c) There are no special acknowledgements needed.

We hope the corrected version is now acceptable in this form.

Thank you for your efforts

Sincerely yours,

Rainer Laskawi